Author’s response to reviews

Title: Understanding Resilience of Female Adolescents towards Teenage Pregnancy: A Crosssectional Survey in Dar es Salaam, Tanzania

Authors:
Constanze Pfeiffer (constanze.pfeiffer@unibas.ch)
Collins Ahorlu (collins.ahorlu@gmail.com)
Sandra Alba (S.Alba@kit.nl)
Brigit Obrist (brigit.obrist@unibas.ch)

Version: 4 Date: 02 Jun 2017

Author’s response to reviews:

All comments raised by the editor were carefully addressed to improve the quality of the paper.

Response to comments:

1. In the discussion section we included a reflection about the higher competence score of pregnant adolescents and young mothers.

2. In order to avoid that resilience and competence are used interchangeably, we revised the terminology in the methods and results section.

3. We separated the results for the indicators: "striving for a good reputation" and "feeling accepted in their community"

4. The definition of terms such as adolescents and young people was included in the Introduction section
5. We clarified the role of the qualitative study component

6. We deleted unclear sentences related to our statistical findings.

7. We included all minor comments